Author’s response to reviews

Title: Identification of a rare 17p13.3 duplication including the BHLHA9 and YWHAE genes in a family with developmental delay and behavioural problems

Authors:

Valeria Capra (valeriacapra@ospedale-gaslini.ge.it)
Marisol Mirabelli-Badenier (marisol.mirabelli@gmail.com)
Michela Stagnaro (stagnaromichela@libero.it)
Andrea Rossi (andrearossi@ospedale-gaslini.ge.it)
Elisa Tassano (eli.tassano@gmail.com)
Stefania Gimelli (stefania.gimelli@gmail.com)
Giorgio Gimelli (giorgio.gimelli@gmail.com)

Version: 4 Date: 10 September 2012

Author’s response to reviews:

Genova, 10/09/2012

Dear Editor,

Attached please find the revised manuscript n. 8281239926995772 entitled “Identification of a rare 17p13.3 duplication including the BHLHA9, YWHAE genes in a family with developmental delay and behavioural problems” by Capra et al., for publication in BMC Medical Genetics.

Reviewer: Damien Bruno

We have accepted all the minor essential revisions and we have corrected the text as suggested. Moreover a native English speaker has revised the manuscript language.

We are grateful to Reviewers for their precious suggestions that allowed us to improve the understanding and the quality of our manuscript.

We are looking forward to hearing from you at your earliest convenience.

My best regards

Giorgio Gimelli